Author's response to reviews

Title: Carotid axillary bypass in a patient with blocked subclavian stents

Authors:

Tarig I Barakat (tarigbarakat@yahoo.co.uk)
Loiuse Kenny (lkenny@doctors.org.uk)
Hazim Khout (hazim.khout@ghnt.nhs.uk)
Grace Timmons (grace.timmons@ghnt.nhs.uk)
Vish Bhattacharya (vbhatta@yahoo.com)

Version: 3 Date: 20 January 2011

Author's response to reviews:

Dear sir/ madam

I have made the changes as requested.

1. The sub-headings in the case presentation section have been removed.
2. The date from figure 3 has been removed.
3. The hospital name in figure 1 has been removed.

Thanks

Tarig Barakat